Author's response to reviews

Title: ATA homozigosity in the IL-10 gene promoter is a risk factor for schizophrenia in Spanish females: a case control study.

Authors:

Berta Almoguera (balmoguera@fjd.es)  
Rosa Riveiro-Alvarez (rriveiro@fjd.es)  
Jorge Lopez-Castroman (jorgecastroman@gmail.com)  
Pedro Dorado (pdorado@unex.es)  
Rosario Lopez-Rodriguez (zeilancharo@hotmail.com)  
Pablo Fernandez-Navarro (pab.fdz@gmail.com)  
Enrique Baca-García (eb2452@columbia.edu)  
Jose Fernandez-Piqueras (ifpiqueras@cbm.uam.es)  
Rafael Dal-Re (rdal-re@intelz.org)  
Francisco Abad-Santos (fabad.hlpr@salud.madrid.org)  
Adrián LLerena (allerena@unex.es)  
Carmen Ayuso (cayuso@fjd.es)  
Spanish Consortium of Pharmacogenetics Research in Schizophrenia SCPRS (cayuso@fjd.es)

Version: 3 Date: 27 May 2011

Author's response to reviews: see over
Dear Sir or Madam:

We enclose the accepted manuscript 225074364754995 - ATA homozigosity in the IL-10 gene promoter is a risk factor for schizophrenia in Spanish females: a case control study with the latest corrections for publication in BMC Medical Genetics.

This is the point-by-point revision of the manuscript. Thank you in advance.

Author groups:

Please include the name of the group within the list of authors, by adding the name to the 'first name' box; and the bracketed abbreviation to the 'last name' box in the submission system. Please remove the names of the members of the group from the title page and place them in the Acknowledgements, as they are to be acknowledged as members of the study group. The groups’ contribution as a whole can be given in the author's contributions section.

We have listed and acknowledge all members of the Spanish Consortium of Pharmacogenetics Research in Schizophrenia in the acknowledge section

Consortia or corporate authors:

These should be consistent between the manuscript and the submission system. Members of a consortium should be listed in the acknowledgements (if anywhere).

Abstract:

Please ensure that you include an abstract in the manuscript file, and that the abstract is identical in the manuscript file and on the submission system. Abstracts should not cite references, nor refer to figures or tables. Please check the instructions for authors to ensure that your abstract follows the correct structure for this journal and article type.

We have included the identical abstract in both places
Authors' information:

Please place the Authors' Contributions section after Competing interests. Please check the instructions for authors on the journal website for the correct format to use for Authors' Contributions.

Author’s Contributions is placed after Competing Interests

Reference citations:

All references should be cited in the text following journal style. Please see the instructions for authors on the journal website for further information.

References have been corrected to the journal style

Tables:

Please ensure that the order in which your tables are cited is the same as the order in which they are provided. Every table must be cited in the text, using Arabic numerals. Please do not use ranges when listing tables.

Tables must not be subdivided, or contain tables within tables. Please note that we are unable to display vertical lines or text within tables, no display merged cells: please re-layout your table without these elements. Tables should be formatted using the Table tool in your word processor. Please ensure the table title is above the table and the legend is below the table. For more information, see the instructions for authors on the journal website.

Tables have been modified according to the journal style